Author's response to reviews

Title: Improvement of pain and regional osteoporotic changes in the foot and ankle by low-dose bisphosphonate therapy for complex regional pain syndrome type I: a case series

Authors:

Yasuhisa Abe (abeyasu@movie.ocn.ne.jp)
Kousuke Iba (iba@sapmed.ac.jp)
Junichi Takada (itakada@kitago-ortho.com)
Takuro Wada (twada@sapmed.ac.jp)
Toshihiko Yamashita (tyamasit@sapmed.ac.jp)

Version: 3 Date: 18 May 2011

Author's response to reviews:

Journal of Medical Case Reports
Editorial Support Agent
Ivan Serraon

Dear Dr. Serraon,

Thank you for your request the changes of our manuscripts. We have corrected the manuscript as followed your suggestion that we should change the title and the case presentation section of the abstract.

Please find enclosed the revised version of our manuscript entitled “Improvement of pain and regional osteoporotic changes in the foot and ankle by low-dose bisphosphonate therapy for complex regional pain syndrome type I: a case series”.

We would appreciate your kind consideration of its publication in Journal of Medical Case Report.

Sincerely Yours,

Kousuke Iba M.D., Ph.D.